Author's response to reviews

Title: Personalising health care: feasibility and future implications

Authors:

Brian Godman (Brian.Godman@ki.se)
Alexander E Finlayson (alexanderfinlayson@gmail.com)
Parneet K Cheema (parneet.cheema@sunnybrook.ca)
Eva Zebedin-Brandl (Eva-Maria.Zebedin@hvb.sozvers.at)
Inaki Gutiérrez-Ibarluzea (osteba7-san@ej-gv.es)
Jan Jones (janjones@nhs.net)
Rickard E Malmstrom (rickard.malmstrom@ki.se)
Elina Asola (elina.asola@stm.fi)
Christoph Baumgärtel (christoph.baumgaertel@ages.at)
Marion Bennie (m.bennie@nhs.net)
Iain Bishop (iain.bishop@nhs.net)
Anna Bucsics (Anna.Bucsics@hvb.sozvers.at)
Eduardo Diogene (ediogene@gencat.cat)
Alessandra Ferrario (A.Ferrario@lse.ac.uk)
Jurij Furst (Jurij.Furst@zzzs.si)
Kristina Garuoliene (krystyna.garuoliene@vlk.lt)
Miguel Gomes (miguelfate@gomes.com)
Katherine Harris (katharine.k.harris@kcl.ac.uk)
Alan Haycox (ahay@liv.ac.uk)
Harald Herholz (harald.herholz@kvhessen.de)
Krystyna Hviding (krystyna.hviding@legemiddelverket.no)
Marija Kalaba (marija.kalaba@rfzo.rs)
Christina Kvalheim (Christina.Kvalheim@legemiddelverket.no)
Ott Laius (Ott.Laius@ravimiamet.ee)
Sven-Ake Lööv (sven-ake.loov@sll.se)
Kamila Malinowska (malinowska@cmkp.edu.pl)
Andrew Martin (andrew.martin8@nhs.net)
Laura McCullagh (lmccullagh@st.james.ie)
Fredrik Nilsson (fredrik.nilsson@tlv.se)
Ken Paterson (ken.paterson@dsl.pipex.com)
Gisbert Selke (Gisbert.Selke@wido.bv.aok.de)
Catherine Sermet (sermet@irdes.fr)
Ulrich Schwabe (ulrich.schwabe@pharma.uni-heidelberg.de)
Dominik Tomek (tdmja@slovak.net.sk)
Durhane Wong-Reiger (durhane@sympatico.ca)
Vera Vlahovic-Palcevski (vvlahovic@inet.hr)
Luka Vonic (luka.vonicina@miz.hr)
Magda Wladysiuk (m.wladysiuk@hta.pl)
Menno Woerkom (m.woerkom@medicijngebruik.nl)
Corrine Zara (czara@catsalut.cat)
Raghib Ali (raghib.ali@ndm.ox.ac.uk)
Author's response to reviews: see over
Dear Ursula and colleagues

Thank you for your recent email containing further comments from the reviewers.

We have re-formatted the revised paper as a 'Review' article including the following (copying the format of Ozomaro U, Wahlestedt C, Nemeroff C. Personalized medicine in psychiatry: problems and promises. Ozomaro et al. BMC Medicine 2013, 11:132 – which we have now included as a reference):

- An abbreviated abstract (175 words)
- Key words after the abstract
- An edited version of the main manuscript – including further editing where deemed necessary to improve the formatting and structure of the paper. In addition, contained in 3 sections of Introduction, Review and Conclusion, with subsections where appropriate. This was followed by a List of Abbreviations, Competing Interests and Financial Disclosure, Author Contributions and References.

In addition, thank you for your email of 13 June. We have now altered the title as well as separated out competing interests and acknowledgements.

Please let is know if you need anything further from us.

We look forward to hearing from you.

Yours faithfully
Dr Brian Godman
Division of Clinical Pharmacology, Karolinska Institutet, Stockholm, Sweden
Mario Negri institute for Pharmacological Research, Milan, Italy
Prescribing Research Group, Liverpool University, UK